**Author's response to reviews**

**Title:** The effects of continuous prostacyclin infusion on regional blood flow and cerebral vasospasm following subarachnoid haemorrhage: study protocol for a randomized controlled trial

**Authors:**
- Rune Rasmussen (rune333@gmail.com)
- Jørn Wetterslev (Wetterslev@ctu.rh.dk)
- Trine Stavngaard (trine.stavngaard@rh.regionh.dk)
- Niels V Olsen (nvolsen@sund.ku.dk)
- Jane Skjøeth-Rasmussen (jane@skjøeth-rasmussen.dk)
- Per-Olof Grande (per-olof.grande@med.lu.se)
- Bertil Romner (bertil.romner@rh.regionh.dk)

**Version:** 4  **Date:** 19 April 2012

**Author's response to reviews:** see over
Dear Editors-in-Chief Doug Altman, Curt Furberg, Jeremy Grimshaw and Peter Rothwell

Thank you for accepting our manuscript for publication in Trials. Attached is our re-revised manuscript with the requested change in title format (highlighted).

Sincerely,

Rune Rasmussen, MD
Dep. of Neurosurgery,
Rigshospitalet, 2090,
Blegdamsvej 9, Copenhagen, DK-2100
Mobile: +45 20157524
E-mail: rune333@gmail.com